Author’s response to reviews

Title: A scoping review of comparisons between abstracts and full reports in primary biomedical research

Authors:

GUOWEI LI (lig28@mcmaster.ca)
Luciana Abbade (lpfabbade@gmail.com)
Yanling Jin (jiny22@mcmaster.ca)
Ikunna Nwosu (nwosui@mcmaster.ca)
Alvin Leenus (lalvin97@hotmail.com)
Muhammad Maaz (maazm@mcmaster.ca)
Mei Wang (wangm59@mcmaster.ca)
Meha Bhatt (bhattm25@mcmaster.ca)
Laura Zielinski (zielinsl@mcmaster.ca)
Nitika Sanger (sangern@mcmaster.ca)
Bianca Bantoto (bantotb@mcmaster.ca)
Candice Luo (luocy2@mcmaster.ca)
Ieta Shams (shamsi@mcmaster.ca)
Hamnah Shahid (shahih4@mcmaster.ca)
Yaping Chang (changy28@mcmaster.ca)
Guangwen Sun (gsun5903@hwdsb.on.ca)
Lawrence Mbuagbaw (mbuagblc@mcmaster.ca)
Zainab Samaan (samaan@mcmaster.ca)
Mitchell Levine (levinem@mcmaster.ca)
Jonathan Adachi (jd.adachi@sympatico.ca)
Author’s response to reviews:

Dear Editorial office,

Regarding the authorship, the third and fourth author should be Ikunna and Yanling respectively; because Yanling was too busy to continue data collection from this January. We have decided Ikunna should be the 3rd author, but somehow I used the incorrect version when submitting.

The original submission should be using all the files dated May 17th; but somehow I used the files dated May 3rd. Therefore the order of authorship (between the 3rd and 4th authors) was incorrectly shown in the system and the WORD files. Very very sorry for the naive mistake we made!

We have uploaded the request of authorship change.

Thanks!

Guowei Li